Author's response to reviews

Title: Determinants of smoking initiation among women in five European countries: a cross-sectional survey

Authors:

Debora L Oh (deboraooh@usc.edu)
Julia E Heck (jeheck@ucla.edu)
Carolyn Dresler (Carolyn.Dresler@arkansas.gov)
Shane Allwright (sllwright@tdc.ie)
Margaretha Haglund (margaretha.haglund@fhi.se)
Sara S Del Mazo (sara.sanchez@ki.se)
Eva Kralikova (eva.kralikova@lf1.cuni.cz)
Isabelle Stucker (stucker@vif.inserm.fr)
Elizabeth Tamang (etamang@ulss15.pd.it)
Ellen R Gritz (egritz@mdanderson.org)
Mia Hashibe (mia.hashibe@utah.edu)

Version: 4 Date: 6 February 2010

Author's response to reviews:

Dear Ms. Abigail Brown,

Thank you for your acceptance of our manuscript. The following formatting changes have been made to the manuscript according to your request:

1. The author list has been updated so that the manuscript matches the online submission system.

2. The abstract has been updated so that the manuscript matches the online submission system.

3. The country has been included in the affiliation details for each author.

4. The footer has been removed from the manuscript.

5. Citation details for reference #30 have been updated; however, we are not certain if this article was e-published ahead of print November 19, 2009, or if the article has in fact been already published.

6. All visible vertical lines have been removed from the tables.

Please let us know if we can be of any further assistance in moving this manuscript forward.

Yours sincerely,

Mia Hashibe